Author's response to reviews

Title: Terrorism in Australia: factors associated with perceived threat and incident-critical behaviours

Authors:

Garry Stevens (g.stevens@uws.edu.au)
Kingsley Agho (k.agho@uws.edu.au)
Melanie Taylor (melanie.taylor@uws.edu.au)
Margo Barr (meyes@doh.health.nsw.gov.au)
Beverley Raphael (b.raphael@uws.edu.au)
Louisa Jorm (louisa.jorm@uws.edu.au)

Version: 8 Date: 18 March 2009

Author's response to reviews:

Line removed from title page "All authors contributed equally to this work" to be consistent with specific contributions that had previously been detailed in Author Contribution section.